Author's response to reviews

Title: Effects of type and level of training on variation in physician knowledge in the use and acquisition of blood cultures: a cross sectional survey

Authors:

Jorge P Parada (jorge.parada@med.va.gov)
David N Schwartz (david.schwartz@hektoen.org)
Gordon D Schiff (gdschiff@aol.com)
Kevin B Weiss (k-weiss@northwestern.edu)

Version: 5 Date: 20 August 2005

Author's response to reviews:

1. I have ensured that the title in the submission system and manuscript are identical
2. I have corrected the formatting of the title in the submission system and manuscript (switched from title case to sentence case)
3. I have switched the reference citations in the manuscript from supr-script to square brackets.
4. I have expanded the competing interests section of the manuscript
5. All authors' names are now included in the references (i.e., names added and et al removed)